Author’s response to reviews

Title: Qualitative Website Analysis of information on Birth after Caesarean Section

Authors:

Valerie L Peddie Mrs (v.peddie@abdn.ac.uk)
Natalie Whitelaw Dr. (nataliewhitelaw1@nhs.net)
Mairead Black Dr. (mairead.black@abdn.ac.uk)
Grant P Cumming Prof. (grant.cumming@nhs.net)
Siladitya Bhattacharya Prof. (s.bhattacharya@abdn.ac.uk)

Version: 4
Date: 19 July 2015

Author’s response to reviews: see over
19\textsuperscript{th} July 2015

The Editor
BMC Pregnancy and Childbirth
BioMed Central
236 Gray's Inn Road
London
WC1X 8HB

Dear Editor,

**MS: 4017138191625667: Qualitative Website Analysis of information on Birth after Caesarean Section**

Thank you for your mail dated 15\textsuperscript{th} July 2015, outlining the editorial points/amendments required to the above manuscript for publication in the *Journal of BMC Pregnancy & Childbirth*. We have now addressed these points, and include details on these below:

1. Please include the section title "Competing Interests" before the competing interests statement in your manuscript.  
   *This has been done*

2. Please confirm that all authors read and approved the final manuscript, including a statement to this effect in the Authors’ Contributions section.  
   *This has been done*

3. Please remove figure file 1 and place table 1 within the main manuscript at the end of the document text, according to the Instructions for Authors.  
   *This has been done*

Kind regards,

Yours sincerely,

Valerie L Peddie
Charge Midwife & Honorary Research Fellow